Author's response to reviews

Title: Patterns of sexually transmitted infections in adolescents and youths in Dar es Salaam, Tanzania

Authors:

Guerino Chalamilla Dr (gchalamilla@muchs-harvard.org)
Judica Mbwana M/s (jmbwana@muchs.ac.tz)
Fred S Mhalu Prof (fmhalu@muchs.ac.tz)
Eunice Mmari Dr (emmari@muchs.ac.tz)
Mtebe Majigo Dr (mmajigo@muchs.ac.tz)
Andrew B Swai Prof (aswai@muchs.ac.tz)
Willy K Urassa Dr (wurassa@muchs.ac.tz)
Eric Sandstrom Prof (eric.sandstrom@karolinska.se)

Version: 2 Date: 17 June 2005

Author's response to reviews: see over
17 June 2005

The Editor
Biomedical Central-Infectious Diseases

Dear Sir/Madam,

RE: RESUBMISSION OF PAPER TITLED: PATTERNS OF SEXUALLY TRANSMITTED INFECTIONS IN ADOLESCENTS AND YOUTH IN DAR ES SALAAM, TANZANIA

I appreciate your recent email regarding the above manuscript which I had submitted earlier titled Patterns of sexually Transmitted infections in Adolescents and youth in Dar es Salaam Tanzania for possible publication in your Journal.

I have changed the structure and contents of the abstract by introducing the subtitles as suggested. So kindly replace the previous abstract with the current one. I have not changed any of the other sections of the manuscript

Best regards,

Yours Sincerely

Willy K Urassa
Senior Lecturer